Author’s response to reviews

Title: Characterizing the Disability Experience among Adults Living with HIV: A Structural Equation Model Using the HIV Disability Questionnaire (HDQ) within the HIV, Health and Rehabilitation Survey

Authors:

Kelly O'Brien (kelly.obrien@utoronto.ca)
Steven Hanna (hannas@mcmaster.ca)
Patricia Solomon (solomon@mcmaster.ca)
Catherine Worthington (worthing@uvic.ca)
Francisco Ibáñez-Carrasco (IbanezCarraF@smh.ca)
Soo Chan Carusone (SChanCarusone@caseyhouse.ca)
Stephanie Nixon (stephanie.nixon@utoronto.ca)
Brenda Merritt (b.merritt@dal.ca)
Jacqueline Gahagan (Jacqueline.Gahagan@Dal.Ca)
Larry Baxter (larrynbaxter@gmail.com)
Patriic Gayle (146iic@gmail.com)
Gregory Robinson (gregoryrobinson8@gmail.com)
Rosalind Baltzer Turje (rbaltzer-turje@drpeter.org)
Stephen Tattle (stattle@rogers.com)
Tammy Yates (tyates@realize.ca)

Version: 1 Date: 15 May 2018

Author’s response to reviews:

Dear Members of the BMC Infectious Diseases Editorial Board,
Thank you for the review of our manuscript. Please find attached a revised manuscript addressing the requested editorial revisions.

We added the following statement to the manuscript text (Methods, recruitment section) as well as the Declarations section.

"Written informed consent was obtained from all participants by individuals answering ‘yes’ to the question ‘I agree to participate in this research study’.

Please don't hesitate to contact me if you require any further information. Thank you for your consideration of our manuscript. We look forward to hearing from you.

Best wishes,

Kelly O'Brien, PhD
Associate Professor
Department of Physical Therapy
University of Toronto
160-500 University Avenue
Toronto, ON  M5G 1V7
Kelly.obrien@utoronto.ca